Author's response to reviews

Title: Analysis of expression profiles of MAGE-A antigens in oral squamous cell carcinoma cell lines

Authors:

Urs DA Müller-Richter (mueller_u2@klinik.uni-wuerzburg.de)  
Albert Dowejko (albert.dowejko@klinik.uni-regensburg.de)  
Tobias Reuther (reuther_t@klinik.uni-wuerzburg.de)  
Johannes Kleinheinz (joklein@uni-muenster.de)  
Torsten E Reichert (torsten.reichert@klinik.uni-regensburg.de)  
Oliver Driemel (oliver.driemel@klinik.uni-regensburg.de)

Version: 2  Date: 2 April 2009

Author's response to reviews:

Dear Professor Kleinheinz,

thank you very much for reviewing my manuscript for Head & Face Medicine. I carefully read the comments of the reviewer and included all proposal in the text. The abstract now includes comments on the NHEK cell line and the implications are discussed in the discussion. A remark on healthy oral mucosa, which does not express MAGE-A antigens in a significant level is also included. I also renumbered the figures. As numbered in the primary submission the figures 1 to 5 and 6 to 11 can grouped together. I hope the manuscript now meets the standards of your journal.

With kind regards

Urs Müller-Richter